Author’s response to reviews

Title: Cognitive appraisal of exposure to specific types of trauma. A study of gender differences

Authors:

Dana HERTA (dana_cristina75@yahoo.com)
Doina COZMAN (doina_octaviancosman@yahoo.com)

Version: 1 Date: 17 Aug 2015

Author’s response to reviews:

Thank you for the feedbacks!

Both issues mentioned in the Editor's letter were addressed.

The manuscript now contains a clear description of the recruitment and assessment procedures, including number of potential candidates, contacted patients and recruited persons.

The manuscript also includes in all its sections a clear engagement with gender in the cultural context in which the study was implemented.